### EUROPEAN CENTRAL BANK

### **WORKING PAPER SERIES**



**WORKING PAPER NO. 290** 

INFLATION PERSISTENCE AND ROBUST MONETARY POLICY DESIGN

**BY GÜNTER COENEN** 

November 2003

#### EUROPEAN CENTRAL BANK

#### **WORKING PAPER SERIES**



### WORKING PAPER NO. 290

# **INFLATION PERSISTENCE** AND ROBUST MONETARY **POLICY DESIGN'**

# **BY GÜNTER COENEN<sup>2</sup>**

# November 2003

e-mail: gunter.coenen@ecb.int.

The paper has benefited from presentations at the meeting of the Society of Computational Economics in Aix-en-Provence, the Annual Congress of the Verein für Socialpolitik in Zürich, the Bank of Finland/CEPR Workshop on "Heterogenous Information and Modeling of Monetary Policy" in Helsinki and a seminar held at the ECB in Frankfurt. Useful comments and suggestions from Ignazio Angeloni, Paul Levine, Juha Kilponen, Frank Smets, Harald Uhlig and Volker Wieland are gratefully acknowledged. The opinions expressed here are those of the author and do not necessarily reflect views of the European Central Bank. Any remaining errors are of course the sole responsibility of the author. This paper can be downloaded without charge from http://www.ecb.int or from the Social Science Research Network electronic library at http://srn.com/labstract\_id=487468.
 Correspondence: Günter Coenen, Directorate General Research, European Central Bank, Kaiserstrasse 29, D-60311 Frankfurt am Main, Germany, phone: +49 69 1344-7887, e-projit untercoenen@ech int.

#### © European Central Bank, 2003

Address	Kaiserstrasse 29
	D-60311 Frankfurt am Main
	Germany
Postal address	Postfach 16 03 19
	D-60066 Frankfurt am Main
	Germany
Telephone	+49 69 1344 0
Internet	http://www.ecb.int
Fax	+49 69 1344 6000
Telex	411 144 ecb d

#### All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged. The views expressed in this paper do not necessarily reflect those of the European Central Bank. The statement of purpose for the ECB Working Paper Series is available form the ECB website, http://www.ecb.int.

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

### Contents

Abs	tract	4
Nor	n-technical summary	5
I	Introduction	7
2	Two models of inflation determination 2.1 The behavioural equations 2.2 Implications for the transmission of monetary policy	2  2  7
3	Evaluating the performance of monetary policy rules 3.1 The methodology 3.2 Optimised benchmark rules	20 21 23
4	The Robustness of Optimised Monetary Policy Rules 4.1 Uncertainty about the forecasting model 4.2 Uncertainty about the rule-generating model	25 26 28
5	Designing robust monetary policy rules	3
6	Sensitivity analysis	35
7	Conclusion	37
Refe	erences	40
Tabl	les	44
Euro	opean Central Bank working paper series	50

#### Abstract

This paper investigates the performance of optimised interest rate rules when there is uncertainty about a key determinant of the monetary transmission mechanism, namely the degree of persistence characterising the inflation process. The paper focuses on the euro area and utilises two variants of an estimated small-scale macroeconomic model featuring distinct types of staggered contracts specifications which induce quite different degrees of inflation persistence. The paper shows that a cautious monetary policy-maker is welladvised to design and implement interest rate policies under the assumption that inflation persistence is high when uncertainty about the prevailing degree of inflation persistence is pervasive.

JEL Classification System: E31, E52, E58, E61

*Keywords*: macroeconomic modelling, staggered contracts, inflation persistence, monetary policy rules, robustness, euro area

#### Non-technical summary

This paper investigates the performance of monetary policy rules when the monetary policymaker is faced with uncertainty about the degree of persistence characterising the inflation process. The degree of inflation persistence represents a key determinant of the monetary transmission mechanism and has important implications for the ability of monetary policy to stabilise inflation relative to output. Hence, monetary policy rules should ideally be designed to perform reasonably well under a range of alternative models of inflation determination which differ with respect to the degree of persistence that they induce.

To examine the consequences of different degrees of inflation persistence for the performance of monetary policy rules, we concentrate on the euro area for which such an examination seems particularly relevant as the single monetary policy of the European Central Bank (ECB) has to focus on the euro area as a whole being a new and relatively unexplored territory. We utilise two variants of the small-scale euro area model developed by Coenen and Wieland (2003) which feature different types of staggered contracts specifications: the nominal wage contracting specification due to Taylor (1980) and the relative real wage contracting specification originally proposed by Buiter and Jewitt (1981) and adapted to empirical work by Fuhrer and Moore (1995). Coenen and Wieland show that both types of specification describe historical euro area data reasonably well.

In terms of methodology, the paper builds on recent work by Levin, Wieland and Williams (1999, 2003) evaluating the performance and robustness of simple monetary policy rules across five different models of the U.S. economy. This methodology involves implementing simple reaction functions describing the response of the short-term nominal interest rate to inflation and the output gap, either observed or forecast, and then optimising over the respective response coefficients. The performance of these optimised interest rate rules is then evaluated with regard to their ability to stabilise inflation and output around their targets, while avoiding undue fluctuations in the nominal interest rate itself.

Unlike the papers by Levin, Wieland and Williams, this paper focuses on one particular determinant of the monetary transmission mechanism, namely the degree of inflation persistence that is induced by alternative models of inflation determination. Thus, while Levin, Wieland and Williams consider a larger set of models which exhibit substantial differences in theoretical specification, in degree of aggregation, in estimation sample and in estimation methodology, we control for all differences except for the different degrees of inflation persistence induced by the two distinct staggered contracts specifications. As a result, any differences in our findings can be attributed to the different degrees of inflation persistence induced by the two staggered contracts specifications.

Following the methodology proposed by Levin, Wieland and Williams, we first evaluate the stabilisation performance of both outcome and forecast-based interest rate rules that are designed for each of the two staggered contracts specifications separately. We then examine the robustness of these optimised interest rate rules in a situation when the monetary policy-maker does not know which of the two staggered contracts specifications represents the "true" model of the inflation process. Specifically, we consider two different scenarios. In the first scenario, the policy-maker is faced with uncertainty regarding the choice of the forecasting model needed to implement a given forecast-based interest rate rule. In the second scenario, the policy-maker faces the even more profound uncertainty as to which of the two contracting specifications to rely upon when designing interest rate rules.

Based on our analysis, we conclude that it may be dangerous to rely too heavily on interest rate rules which are implemented and/or designed under the assumption that the degree of inflation persistence is low. Following the prescriptions of such rules may result in disastrous stabilisation outcomes if the true inflation process turns out to be much more persistent. In contrast, rules which are implemented and/or designed under the assumption that the degree of inflation persistence is relatively high also perform reasonably well if inflation is considerably less persistent. Hence, a cautious monetary policy-maker is welladvised to take monetary policy decision under the assumption that the inflation process is characterised by a high degree of persistence until strong evidence in favour of a lowpersistence regime has emerged. In this context, we illustrate that using pooled forecasts rather than relying on a single model's inflation forecast can serve as a simple means to insure the policy-maker at least against the risks arising from the use of the wrong forecasting model, regardless of the prevailing degree of inflation persistence.

Finally, we identify the key operating characteristics of simple interest rate rules that are robust to the different degrees of inflation persistence induced by the two staggered contracts specifications under investigation. To this end, we optimise the simple interest rate rules across the alternative staggered contracts specifications simultaneously. Our results confirm previous findings by Levin, Wieland and Williams: robust rules respond to inflation and output gap forecasts with horizons that do not extend too far into the future and also incorporate a substantial degree of interest rate inertia. Specifically, we find that a first-difference rule which relates changes in the short-term nominal interest rate to the one-year ahead forecast of inflation and the current output gap already goes a long way towards making monetary policy robust to the different degrees of inflation persistence that are induced by the two types of staggered contracts specifications. To the extent that such a rule performs remarkably well under both staggered contracts specifications and in light of the large uncertainty about how to model euro area inflation, we tentatively conclude that such a rule may serve as a useful benchmark for model-based evaluations of monetary policy in the euro area.

### 1 Introduction

There is an active and rapidly growing literature on the evaluation of structural models of inflation determination.<sup>1</sup> While theoretical models, starting with the staggered contracts models of Taylor (1980) and Calvo (1983), propose that current inflation depends on future inflation and a measure of current excess demand, recent empirical research has highlighted that these models, in their simplest specification, typically fail to explain the degree of inflation persistence observed in the data. This failure has been documented most prominently by Fuhrer and Moore (1995) who rejected Taylor's nominal wage contracting model for U.S. inflation data and found strong empirical evidence in favour of the relative real wage contracting model originally proposed by Buiter and Jewitt (1981) which induces a considerably higher degree of inflation persistence.

More recently, Taylor (2000) and Cogley and Sargent (2001) have observed that the degree of persistence in U.S. inflation has been drifting downward in the 1980s and 1990s as inflation has come under control.<sup>2</sup> Taylor (2000) suggests that the diminished degree of inflation persistence may be due to changes in the orientation of monetary policy. In a low-inflation regime with credible monetary policy, inflation expectations may become contained and, hence, price and wage setters may be less inclined to change their contracts in response to shocks. Similarly, Brayton, Roberts and Williams (1999) argue that globalisation has increased competition in the products markets, thereby squeezing mark-ups and yielding reductions in prices. Although Staiger, Stock and Watson (2001) do not find empirical evidence in favour of such theories that place considerable weight on changes in price and wage-setting behaviour when revisiting U.S. wage and price inflation in the 1990s, more favourable evidence may emerge as data from the low-inflation regime accumulate.

In the light of the ongoing controversy about the appropriate specification of structural

<sup>&</sup>lt;sup>1</sup>See for example the evaluations by Galí and Gertler (1999), Sbordone (2002) and Rudd and Whelan (2002) for the United States and the studies by Galí, Gertler and López-Salido (2001) and Coenen and Wieland (2003) for the euro area.

 $<sup>^{2}</sup>$ In line with this observation, recent work by Guerrieri (2002) suggests that even Taylor-style contracts are not rejected for U.S. data once the estimation period is extended to the more recent past.

models of inflation determination and more recent indications that the law of motion for inflation may have altered, this paper investigates the performance of simple monetary policy rules when the monetary policy-maker is faced with uncertainty about the degree of persistence characterising the inflation process. The degree of inflation persistence represents a key determinant of the monetary transmission mechanism and has important implications for the ability of monetary policy to stabilise inflation relative to output. Hence, monetary policy rules should ideally be designed to perform reasonably well under a range of alternative models of inflation determination which differ with respect to the degree of inflation persistence that they induce.

To examine the consequences of different degrees of inflation persistence for the performance of monetary policy rules, we concentrate on the euro area for which such an examination seems particularly relevant as the single monetary policy of the European Central Bank (ECB) has to focus on the euro area as a whole being a new and relatively unexplored territory. We utilise two variants of the small-scale euro area model developed by Coenen and Wieland (2003) which feature different types of staggered contracts specifications: the nominal wage contracting specification due to Taylor (1980) and the relative real wage contracting specification originally proposed by Buiter and Jewitt (1981) and adapted to empirical work by Fuhrer and Moore (1995).<sup>3</sup> Both types of contracting specification are found to describe historical euro area data reasonably well.

Comparing the euro area results to those obtained for France, Germany and Italy separately, Coenen and Wieland (2003) show that the relative real wage contracting specification does quite well in countries which transitioned out of a high inflation regime such as France and Italy, while the nominal wage contracting specification describes German data better

<sup>&</sup>lt;sup>3</sup>Thus, the estimated small-scale euro area model belongs to the class of New-Keynesian models which have gained increased popularity in macroeconomic modelling over the recent years. This class of models includes most of the smaller-scale models currently used for research on monetary policy (see for example the backward-looking model of Rudebusch and Svensson (1999), the models with rational expectations and nominal rigidities of Fuhrer and Moore (1995), Fuhrer (1997) and Orphanides and Wieland (1998)) and the models with optimising agents of Rotemberg and Woodford (1997, 1999) and McCallum and Nelson (1999)), as well as large-scale policy models such as the Federal Reserve Board's FRB/U.S. model (see Brayton and Tinsley (1996)), the ECB's Area-Wide Model (see Fagan, Henry and Mestre (2001)) or the multi-country model of Taylor (1993a).

which exhibit a substantially lower degree of inflation persistence. This finding may be attributed to different degrees of nominal rigidity in the price and wage-setting behaviour across economies, but it may also reflect different degrees of credibility of the respective monetary regimes over the estimation period. The estimation results with German data also provide indirect empirical support for the thesis that the degree of inflation persistence is lower in a stable monetary regime with low average inflation, because of the reduced pricing power of firms as suggested by Taylor (2000).

Thus, as far as the future of the European Monetary Union (EMU) is concerned, the estimation based on historical euro area data may overstate the case for the relative real wage contracting model. In this case, nominal rigidities à la Taylor may provide a better description of the inflation process than Fuhrer-Moore-type rigidities and, in terms of evaluating alternative monetary policy strategies, a policy-maker who is optimistic about the output losses associated with stabilising inflation may prefer to use the nominal wage contracting specification, while a pessimist may prefer the relative real wage contracting specification. Given the high degree of uncertainty about the determination of euro area inflation in the future, however, a robust monetary policy strategy for the euro area should perform reasonably well under both contracting specifications.<sup>4</sup>

In terms of methodology, this paper builds on recent work by Levin, Wieland and Williams (1999, 2003) – henceforth referred to as LWW (1999, 2003) – evaluating the performance and robustness of simple monetary policy rules across five different models of the U.S. economy.<sup>5</sup> This methodology involves implementing simple reaction functions describing the response of the short-term nominal interest rate to inflation and the output gap, either observed or forecast, and then optimising over the respective response coefficients.

<sup>&</sup>lt;sup>4</sup>Another hypothesis, though with similar consequences, is that due to heterogeneity in the persistence of the national inflation rates in the countries that form the countries of the euro area, the use of aggregated euro area inflation data induces an upward bias in the estimated degree of inflation persistence. The latter would be an empirical artefact and therefore considered misleading as regards the evaluation of alternative monetary policy strategies.

<sup>&</sup>lt;sup>5</sup>There are alternative approaches to analysing the consequences of uncertainty about the structure of the economy (see for example Giannoni (2002), Hansen and Sargent (2002), Onatski and Stock (2002), Onatski and Williams (2003) and Tetlow and von zur Muehlen (2001)). This coexistence of alternative approaches reflects that there has not yet emerged a consensus on how to address the issue of model uncertainty.

The performance of these optimised interest rate rules is then evaluated with regard to their ability to stabilise inflation and output around their targets, while avoiding undue fluctuations in the nominal interest rate itself.<sup>6</sup>

Unlike the papers by LWW, this paper focuses on one particular determinant of the monetary transmission mechanism, namely the degree of inflation persistence that is induced by alternative models of inflation determination. Thus, while LWW consider a larger set of models which exhibit substantial differences in theoretical specification, in degree of aggregation, in estimation sample and in estimation methodology, we control for all differences except for the different degrees of inflation persistence induced by the two distinct staggered contracts specifications. As a result, any differences in our findings can be attributed to the different degrees of inflation persistence induced by the two contracting specifications.<sup>7</sup>

We start our analysis by comparing the characteristics of inflation and output gap dynamics under the two staggered contracts specifications with an empirical benchmark policy rule imposed. To this end, we report the responses of inflation and the output gap to an unexpected tightening of monetary policy to illustrate the differences in the transmission of monetary policy. Our subsequent analysis proceeds in two steps. Following the methodology proposed by LWW, we first evaluate the stabilisation performance of both outcome and forecast-based interest rate rules which are designed for each of the two staggered contracts specifications separately. We then examine the robustness of these optimised interest rate rules in a situation when the monetary policy-maker does not know which of the two staggered contracts specifications represents the "true" model of the inflation process. Specifically, we consider two different scenarios. In the first scenario, the policy-maker is faced with uncertainty regarding the choice of the forecasting model needed to implement a given forecast-based interest rate rule. In the second scenario, the policy-maker faces the

 $<sup>^{6}</sup>$ Earlier studies of the performance of interest rate rules across a range of macroeconomic models of the U.S. economy are provided in Bryant, Hooper and Mann (1993) and Taylor (1999).

<sup>&</sup>lt;sup>7</sup>Our study is related to recent work by Jääskelä (2002) on the implications of inflation persistence for the design of optimal monetary policy. However, while we evaluate the performance of both outcome and forecast-based interest rate rules in an empirically estimated model of the euro area with two distinct supply-side specifications, Jääskelä focuses on the performance of outcome-based interest rate rules in a highly-stylised calibrated model with a hybrid Phillips curve.

even more profound uncertainty as to which of the two contracting specifications to rely upon when designing interest rate rules.

Based on our analysis, we conclude that it may be dangerous to rely too heavily on interest rate rules which are implemented and/or designed under the assumption that the degree of inflation persistence is low. Following the prescriptions of such rules may result in disastrous stabilisation outcomes if the true inflation process turns out to be much more persistent. In contrast, rules which are implemented and/or designed under the assumption that the degree of inflation persistence is relatively high also perform reasonably well if inflation is considerably less persistent. Hence, a cautious monetary policy-maker is welladvised to take monetary policy decision under the assumption that the inflation process is characterised by a high degree of persistence until strong evidence in favour of a lowpersistence regime has emerged. In this context, we illustrate that using pooled forecasts rather than relying on a single model's inflation forecast can serve as a simple means to insure the policy-maker against risks arising from the use of the wrong forecasting model, regardless of the prevailing degree of inflation persistence.

Finally, we identify the key operating characteristics of simple interest rate rules that are robust to the different degrees of inflation persistence induced by the two staggered contracts specifications. To this end, we optimise the simple interest rate rules across the alternative staggered contracts specifications simultaneously. Our results confirm previous findings by LWW: robust rules respond to inflation and output gap forecasts with horizons that do not extend too far into the future and also incorporate a substantial degree of interest rate inertia. Specifically, we find that a first-difference rule which relates changes in the short-term nominal interest rate to the one-year ahead forecast of inflation and the current output gap already goes a long way towards making monetary policy robust to the different degrees of inflation persistence that are induced by the two types of staggered contracts specifications. To the extent that such a rule performs remarkably well under both types of staggered contracts specifications and in the light of the large uncertainty about how to model euro area inflation, we tentatively conclude that such a rule may serve as a useful benchmark for model-based evaluations of monetary policy in the euro area.

The remainder of this paper is organised as follows. Section 2 outlines the behavioural equations of the euro area model with the two distinct staggered contracts specifications and illustrates the implied differences in inflation and output gap dynamics under an estimated benchmark rule. Section 3 briefly describes the methodology used for evaluating the performance of simple interest rate rules and provides a set of optimised benchmark rules for each of the two staggered contracts models. Section 4 evaluates the robustness of these optimised rules when there is uncertainty about the forecasting and/or the rule-generating model, while Section 5 identifies the operating characteristics of simple interest rate rules that are robust to different degrees of inflation persistence. Section 6 reports additional sensitivity analysis and Section 7 concludes.

## 2 Two Models of Inflation Determination

To analyse the robustness of monetary policy rules when there is uncertainty about the degree of inflation persistence, we utilise two variants of the small-scale euro area model developed by Coenen and Wieland (2003). The first variant employs the nominal wage contracting specification due to Taylor (1980), and the second the relative real wage contract specification originally proposed by Buiter and Jewitt (1981) and adapted to empirical work by Fuhrer and Moore (1995). These two contracting specifications differ with respect to the degree of inflation persistence that they induce, because relative real wage contracts give more weight to past inflation.<sup>8</sup>

#### 2.1 The Behavioural Equations

The behavioural equations of the small-scale euro area model are indicated in **Table 1**. As shown in model equation (M-1), the aggregate price level  $p_t$  is determined as a weighted

<sup>&</sup>lt;sup>8</sup>There are other mechanisms which have been proposed in the literature as a means to induce lagdependent inflation dynamics. For example, Galí and Gertler (1999) allow for a fraction of backwardlooking firms in the staggered nominal contracts model of Calvo (1983), which are assumed to follow a "rule of thumb" when changing prices, while Christiano, Eichenbaum and Evans (2001) assume that nominal contracts are indexed to past prices.

average of staggered nominal wage contracts signed over the past year,  $x_{t-i}$  (i = 0, 1, 2, 3), which are still in effect in period t.<sup>9</sup> Following Fuhrer and Moore (1995), the weights  $f_i$  on contract wages from different periods are assumed to be a downward-sloping linear function of contract length. This function depends on a single parameter, the slope s.

Price level	$p_t = f_0 x_t + f_1 x_{t-1} + f_2 x_{t-2} + f_3 x_{t-3},$ where $f_i = 0.25 + (1.5 - i) s, s \in (0, 1/6]$	(M-1)
Contract wage		
a) Taylor	$x_t = \mathbf{E}_t \left[ \sum_{i=0}^3 f_i  p_{t+i} + \gamma \sum_{i=0}^3 f_i  y_{t+i} \right] + \epsilon_t^x$	(M-2a)
b) Fuhrer-Moore	$x_{t} - p_{t} = \mathbf{E}_{t} \left[ \sum_{i=0}^{3} f_{i} v_{t+i} + \gamma \sum_{i=0}^{3} f_{i} y_{t+i} \right] + \epsilon_{t}^{x},$ where $v_{t} = \sum_{i=0}^{3} f_{i} \left( x_{t-i} - p_{t-i} \right)$	(M-2b)
Aggregate demand	$y_{t} = \delta_{1} y_{t-1} + \delta_{2} y_{t-2} + \phi \left( r_{t-1}^{l} - r^{*} \right) + \epsilon_{t}^{d},$ where $y_{t} = q_{t} - q_{t}^{*}$	(M-3)
Term structure	$i_t^l = \mathcal{E}_t \left[ \frac{1}{8} \sum_{i=0}^7 i_{t+i}^s \right]$	(M-4)
Fisher equation	$r_{t}^{l} = i_{t}^{l} - \mathcal{E}_{t} \left[ \frac{1}{8} \sum_{i=1}^{8} \pi_{t+i} \right],$ where $\pi_{t} = 4 \left( p_{t} - p_{t-1} \right)$	(M-5)

Table 1: A Small-Scale Euro Area Model with Staggered Wage Contracts

Notes: p: aggregate price level; x: nominal contract wage; y: output gap;  $\epsilon^x$ : contract wage shock; v: real contract wage index;  $r^l$ : long-term real interest rate;  $r^*$ : equilibrium real interest rate;  $\epsilon^d$ : aggregate demand shock; q: actual output;  $q^*$ : potential output;  $i^l$ : long-term nominal interest rate;  $i^s$ : short-term nominal interest rate;  $\pi$ : one-quarter inflation. Prices, wages and output are expressed in logarithmic form, and interest rates and inflation are expressed at annualised rates.

The staggered contracts models of Taylor and Fuhrer-Moore induce nominal rigidities, because workers negotiate long-term contracts and compare the contract wage to past contracts that are still in effect and future contracts that will be negotiated over the life of this

<sup>&</sup>lt;sup>9</sup>Thus, like Fuhrer and Moore (1995), we treat the aggregate price and aggregate wage indices interchangeably, which is consistent with a fixed mark-up. For recent studies considering wage and price stickiness separately, see Taylor (1993a), Erceg, Henderson and Levin (2000) and Amato and Laubach (2000).

contract. As a result only a subset of nominal wage contracts are adjustable at a given point in time. The distinction between Taylor and Fuhrer-Moore-type wage contracts concerns the definition of the wage indices that form the basis of this comparison.

Under Taylor's specification defined by equation (M-2a), the nominal wage contract  $x_t$ is negotiated with reference to the price level that is expected to prevail over the life of the contract,  $p_{t+i}$ , as well as the expected output gap over this period,  $y_{t+i}$ . The operator  $E_t[\cdot]$  indicates the model-consistent expectation of a particular variable conditional on all information available in period t.<sup>10</sup> Since the price indices  $p_{t+i}$  reflect contemporaneous and preceding contract wages, (M-2a) implies that wage setters look at an average of nominal contract wages negotiated in the recent past and expected to be negotiated in the near future when setting the current contract wage. In other words, they take into account nominal wages that apply to overlapping contracts. If wage setters expect the output gap to be positive,  $y_{t+i} > 0$ , they adjust the current contract wage upwards relative to overlapping contracts. The sensitivity of contract wages to the output gap is measured by  $\gamma$ . The contract wage shock  $\epsilon_t^x$  is assumed to be serially uncorrelated.

Under the Fuhrer-Moore specification defined by equation (M-2b), workers negotiating their nominal wage compare the implied real wage expected to prevail over the life of their contract with the real wages on overlapping contracts in the recent past and near future. This specification implies that the expected real wage under contracts signed in the current period is set with reference to an average of real contract wage indices expected to prevail over the current and the next three quarters,  $v_{t+i}$ .<sup>11</sup> Thus, the Fuhrer-Moore contracts should not be understood as reflecting real wage rigidity, but rather as representing an alternative nominal rigidity.

Equations (M-1) and (M-2a, M-2b) represent rules for price and contract wage-setting

 $<sup>^{10}</sup>$ We employ the AIM algorithm of Anderson and Moore (1985), which uses the Blanchard and Kahn (1980) method for solving linear rational expectations models, to compute model-consistent expectations.

<sup>&</sup>lt;sup>11</sup>Here we follow Fuhrer and Moore (1995) and use the current price level in the definition of the real contract wage instead of using the average price level expected to prevail over the life of the contract, which would be theoretically preferable. For a more detailed discussion of variations of relative real wage contracts see Coenen and Wieland (2003).

that are not explicitly derived from a framework with optimising agents. However, they need not necessarily be inconsistent with such a framework. More recently, Taylor-style staggered contracts have been analysed within more fully fleshed-out dynamic general equilibrium models (see for example Chari, Kehoe and McGrattan (2000) or King and Wolman (1999)). Starting with a representative agent model with monopolistically competitive firms these studies add the constraint that *prices*, rather than *wages* are set in a staggered fashion for a fixed number of periods. A log-linear approximation of a stripped-down version of these equilibrium models then implies a contract price equation that coincides with Taylor's contract wage equation (M-2a) with the parameter  $\gamma$  being a function of deeper technology and preference parameters. The Fuhrer-Moore contracting model, however, has typically been criticised for lacking such microeconomic foundations.

To complete our macroeconomic model of the euro area, it remains to specify aggregate demand and the transmission of monetary policy. As regards the determination of aggregate demand, equation (M-3) relates the output gap, i.e. the deviation of actual output from its potential,  $y_t = q_t - q_t^*$ , to two lags of itself and to the lagged ex-ante long-term real interest rate,  $r_{t-1}^l$ . Since our analysis is focused on the implications of different types of staggered contracts specifications we assume for simplicity that potential output  $q_t^*$  is exogenous. In the short run, actual output  $q_t$  may deviate from its long-run potential due to the nominal rigidities arising from the staggering of contracts. The demand shock  $\epsilon_t^d$  is assumed to be serially uncorrelated. The rationale for including lags of the output gap is to account for habit formation in consumption as well as adjustment costs and accelerator effects in investment. We use the lagged instead of the contemporaneous value of the long-term real interest rate to allow for a transmission lag of monetary policy.

Two equations relate the long-term real interest rate  $r_t^l$  to the short-term nominal interest rate  $i_t^s$  which is assumed to be the principal instrument of monetary policy. First, as to the determination of the long-term nominal interest rate  $i_t^l$  defined by equation (M-4), we rely on the accumulated forecasts of the short-term interest rate over two years. These accumulated forecasts will coincide with the long-term interest rate forecast for this horizon under the expectations hypothesis of the term structure. The term premium is assumed to be constant and equal to zero. And second, according to the Fisher relation defined by equation (M-5), we obtain the ex-ante long-term real interest rate  $r_t^l$  by subtracting inflation expectations over the following two years, with  $\pi_{t+i} = 4 (p_{t+i} - p_{t+i-1})$  denoting the annualised one-quarter inflation rate.

Aggregate supply $^{(a)}$	8	$\gamma$		p-value <sup>(c)</sup>
a) Taylor	$0.0456 \\ (0.0465)$	$\begin{array}{c} 0.0115 \ (0.0053) \end{array}$		0.3186[2]
b) Fuhrer-Moore	$\begin{array}{c} 0.0742 \\ (0.0245) \end{array}$	$\begin{array}{c} 0.0212 \\ (0.0048) \end{array}$		0.2602 [2]
Aggregate demand $^{(b)}$	$\delta_1$	$\delta_2$	$\phi$	p-value <sup>(c)</sup>
	1.1807 (0.1006)	-0.2045 (0.1065)	-0.0947 (0.0333)	0.2307[5]

Table 2: The Parameter Estimates of the Small-Scale Euro Area Model

Notes: <sup>(a)</sup> Simulation-based indirect estimates using a VAR(3) model for the annualised one-quarter inflation rate and the output gap as auxiliary model. Standard errors in parentheses. <sup>(b)</sup> GMM estimates using a constant, lagged values of the output gap, the annualised one-quarter inflation rate and the short-term nominal interest rate as instruments. Standard errors in parentheses. <sup>(c)</sup> Probability value associated with the test of overidentifying restrictions. Number of overidentifying restrictions in brackets.

Estimates of the model's parameters are taken from Coenen and Wieland (2003) and summarised in **Table 2**. The upper panel of the table shows the estimated parameters of the alternative price and wage setting specifications that form the supply side of the model. As indicated by the p-values for the tests of overidentifying restrictions that were imposed when estimating the staggered contracts specifications, neither Taylor nor Fuhrer-Mooretype contracts can be rejected on statistical grounds. Taylor-type contracts, however, are favoured somewhat by a higher p-value. The lower panel of the table shows the estimated parameters of the aggregate demand equation. The coefficients on the two lags of the output gap exhibit an accelerator pattern and the interest rate sensitivity of aggregate demand is sizeable. For further details on the estimation of the model's supply-side specifications and its demand side we refer the reader to Coenen and Wieland (2003).

#### 2.2 Implications for the Transmission of Monetary Policy

In principle, the estimated staggered contracts specifications together with the estimated aggregate demand equation would be sufficient to evaluate the stabilisation performance and robustness of alternative monetary policies. However, if we wish to know how the different contracting specifications would have affected the transmission of monetary policy historically we also need to specify an empirical benchmark. We could do so by estimating a policy reaction function which captures the historical path of the euro area short-term nominal interest rate.

Since (GDP-) weighted averages of European interest rates preceding the formation of European Monetary Union in 1999 seem unlikely to be appropriate as a measure of the euro area-wide historical monetary policy stance, however, we resort to estimating a reaction function for the German interest rate (that we already used in estimating the aggregate demand equation discussed above). After all, movements in German interest rates eventually had to be mirrored by the other European countries to the extent that they intended to maintain exchange rate parities within the European Monetary System (EMS).<sup>12</sup>

The benchmark rule that we estimate is a forecast-based interest rate rule which relates short-term nominal interest rates to variations of the one-year-ahead forecast of annual inflation in deviation from the policy-maker's target  $\pi^*$  and the current output gap and also allows for interest rate inertia, i.e. "smoothing".<sup>13</sup> We estimate a generalised form of such a forward-looking rule allowing for up to two lags of the short-term nominal interest

<sup>&</sup>lt;sup>12</sup>Clarida, Galí and Gertler (1998) also argue that German monetary policy had a strong influence on interest rate policy in the U.K., France and Italy throughout the EMS period. More recently, Faust, Rogers and Wright (2001) estimate an interest rate reaction function for Germany and use it to predict the interest rate the ECB would be setting, were it to behave like the Bundesbank.

<sup>&</sup>lt;sup>13</sup>Work by Clarida and Gertler (1997) and Clarida, Galí and Gertler (1998) suggests that German interest rate policy since 1979 is summarised quite well by such a forecast-based interest rate rule.

rate using quarterly German data (standard errors in parentheses):

$$i_{t}^{s} = 1.0670 \ i_{t-1}^{s} - 0.2764 \ i_{t-2}^{s} + (1 - 1.0670 + 0.2764) \left( r^{*} + E_{t}[\tilde{\pi}_{t+4}] \right)$$

$$(0.0271) \qquad (0.0320) + 0.1778 \left( E_{t}[\tilde{\pi}_{t+4}] - \pi^{*} \right) + 0.0388 \ y_{t} + \epsilon_{t}^{i^{s}}$$

$$(0.0241) \qquad (0.0169)$$

Here,  $i_t^s$  corresponds to the three-month money market rate, the annual inflation rate  $\tilde{\pi}_t = p_t - p_{t-4}$  is the annual change in the log-level of the GDP deflator and the output gap measure  $y_t$  has been constructed using quarterly real GDP data and annual output gap estimates reported in OECD (2002).  $r^*$  again denotes the equilibrium real interest rate and  $\pi^*$  reflects the monetary policy-maker's inflation target. The term  $\epsilon_t^{i^s}$  captures unexpected shocks to monetary policy. Motivated by earlier work of Clarida, Galí and Gertler (1998), the estimation period was chosen to start in the second quarter of 1979 with the formation of the European Monetary System; it ends in the fourth quarter of 1998, prior to the launch of the euro in January 1999.<sup>14</sup>

The estimated short-run response coefficients in the interest rate rule capture the pattern of stabilisation policy during the 1980s and 1990s in Germany. Not surprisingly, the coefficient on the one-year-ahead inflation forecast is sizeable, implying a long-run coefficient of about 1.85. This ensures determinacy and stability of the rational expectations equilibrium, when solving the model under either of the wage contracting specifications.<sup>15</sup> The coefficient on the current output gap is a good bit smaller but positive; and the estimated degree of interest rate smoothing turns out to be relatively high.

In the deterministic steady state of the model the output gap is zero and the short and real interest rate equal its equilibrium value  $r^*$ . Since the alternative staggered contracts specifications do not impose any restriction on the steady-state inflation rate, the latter is

<sup>&</sup>lt;sup>14</sup>The interest rate rule has been estimated by the Generalised Method of Moments (GMM) using lags up to order three of the interest rate, one-quarter inflation and the output gap as instruments. In addition, the current value and lags up to order two of the ratio of government expenditure to potential output have been used to account for demand pressures in the aftermath of German reunification which were at least partly due to fiscal measures.

<sup>&</sup>lt;sup>15</sup>See Woodford (2000) for a detailed discussion of the conditions regarding the size of the interest rate response coefficients on current or expected inflation in order to guarantee uniqueness of the rational expectations equilibrium in a standard New-Keynesian sticky-price model.

determined by monetary policy alone and equals the policy-maker's target rate  $\pi^*$  in the interest rate rule.



Figure 1: Responses to an Unexpected Policy Tightening (50 Basis Points)

**Figure 1** provides a comparison of the effects of an unexpected tightening of monetary policy by 50 basis points under the two different wage contracting specifications. The upper two panels depict the responses of annual inflation and the output gap, while the lower two panels show the responses of the short-term nominal interest rate and the ex-ante long-term real interest rate. The dashed lines refer to the responses under Taylor's wage contracting

specification, while the dash-dotted lines correspond to the responses under the contracting specification due to Fuhrer and Moore. Both the equilibrium real interest rate and the policy-maker's inflation target have been normalised to zero.

Qualitatively, the tightening of policy has the same consequences under the two wage contracting specifications. As the nominal interest rate rises unexpectedly, demand falls short of potential and inflation falls below target, with the dynamic adjustment being drawn out lastingly. Quantitatively, however, the responses exhibit some noticeable differences. Most importantly, the disinflation effect is considerably larger under Fuhrer-Moore-type contracts with the timing of the peak effect on inflation noticeably delayed relative to that on demand. By contrast, the decline in inflation is less pronounced under Taylor-type contracts and the timing of the peak effects on inflation and output is almost identical. These differences reflect that expectations regarding future inflation drop more sharply and are much more persistent under Fuhrer-Moore contracts than under Taylor contracts, as discussed in more detail in Coenen and Wieland (2003). With the nominal interest rate being set in response to expectations of future inflation, the empirical benchmark rule prescribes to lower the nominal interest rates under Fuhrer-Moore contracts more decisively than under Taylor contracts. As a result, ex-ante long-term real interest rates rise more strongly under the latter, inducing a more pronounced decline in demand.

Based on the documented patterns of the impulse response functions, we summarise that a given monetary policy rule may perform quite differently in terms of inflation and output gap stabilisation, depending on the type of staggered contracts. Hence, it is evident why monetary policy-makers should be concerned about the model of inflation determination when designing policies.

## 3 Evaluating the Performance of Monetary Policy Rules

We now proceed to describe the methodology which we will use to evaluate the stabilisation performance of alternative monetary policies in the presence of uncertainty about the true model of inflation determination. Our starting point is an evaluation of simple interest rate rules which respond to outcomes or forecasts of annual inflation and the output gap and allow for inertia due to dependence on the lagged short-term nominal interest rate.

#### 3.1 The Methodology

Following the approach in LWW (2003), we consider a parametric family of simple interest rate rules,

$$i_{t}^{s} = \rho i_{t-1}^{s} + (1-\rho) \left( r^{*} + \mathbf{E}_{t} \left[ \tilde{\pi}_{t+\theta} \right] \right) + \alpha \mathbf{E}_{t} \left[ \tilde{\pi}_{t+\theta} - \pi^{*} \right] + \beta \mathbf{E}_{t} \left[ y_{t+\kappa} \right],$$

where again  $i_t^s$  denotes the short-term nominal interest rate,  $r^*$  is the equilibrium real interest rate,  $\tilde{\pi}_t = p_t - p_{t-4}$  is the annual inflation rate,  $\pi^*$  denotes the inflation target, and  $y_t$  is the output gap. Under rational expectations, the operator  $E_t[\cdot]$  indicates the model-consistent forecast of a particular variable, using information available in period t. The parameters  $\theta$  and  $\kappa$  denote the length of the forecast horizons for inflation and the output gap respectively. This specification accommodates both forecast-based rules (with forecast horizons  $\theta, \kappa > 0$ ) and outcome-based rules ( $\theta = \kappa = 0$ ) and simplifies to the one proposed by Taylor (1993b) if  $\theta = \kappa = 0$  and  $\rho = 0$ . For fixed inflation and output gap forecast horizons  $\theta$  and  $\kappa$ , the above family of policy rules is defined by the triplet of response coefficients  $\rho$ ,  $\alpha$  and  $\beta$ .

In our evaluation of the stabilisation performance of particular versions of the parametric family of policy rules, we assume that the policy-maker has a standard loss function equal to the weighted sum of the unconditional variances of inflation, the output gap and changes in the short-term nominal interest rate,

$$\mathcal{L} = \operatorname{Var}[\pi_t] + \lambda \operatorname{Var}[y_t] + \mu \operatorname{Var}[\Delta i_t^s].$$

Here, inflation is measured by the annualised one-quarter inflation rate,  $\pi_t = 4 (p_t - p_{t-1})$ . The weight  $\lambda \geq 0$  refers to the policy-maker's preference for reducing output variability relative to inflation variability, and the weight  $\mu \geq 0$  on the variability of changes in the short-term nominal interest rate,  $\Delta i_t^s = i_t^s - i_{t-1}^s$ , reflects a desire to avoid undue fluctuations in the nominal interest rate itself. Establishing this loss function is consistent with the assumption that the policy-maker aims at stabilising inflation around the inflation target  $\pi^*$  and actual output around potential, with the concern regarding excessive interest rate variability justified by financial stability considerations or the risk of hitting the zero-interest-rate bound.<sup>16,17</sup>

For fixed inflation and output gap forecast horizons  $\theta$  and  $\kappa$ , the parametric family of interest rate rules defined above is optimised by minimising the policy-maker's loss function  $\mathcal{L}$  with respect to the triplet of coefficients  $\rho$ ,  $\alpha$  and  $\beta$ . In this context, in order to evaluate the policy-maker's loss function, we repeatedly need to compute the unconditional variances of the model's endogenous variables for a particular interest rate rule. In preparation for these computations, we first identify the series of historical structural shocks that would be consistent with the alternative contracting specifications under rational expectations with the estimated benchmark for historical monetary policy imposed.<sup>18</sup> Based on the covariance matrix of the structural shocks, it is then possible to calculate the unconditional covariance matrix of the endogenous variables for a given interest rate rule by applying standard methods to the reduced-form solution of the model including that rule.

In the subsequent analysis, we will consider four alternative values for the relative weight on output gap variability, namely  $\lambda = 0, 1/2, 1, 2$ . Regarding the weight on the variability of interest rate changes, we concentrate the analysis on a fixed value of  $\mu = 1$ . This weight is relatively high, but avoids extreme and counterfactual interest rate variability under the optimised rules. In the sensitivity analysis at the end of the paper we report additional results for a lower weight of  $\mu = 0.1$  on interest rate variability. There, we will also briefly

<sup>&</sup>lt;sup>16</sup>For an explicit derivation of the policy-maker's loss function  $\mathcal{L}$  from quadratic intertemporal preferences the reader is referred to Rudebusch and Svensson (1999). In Svensson's (1999) terminology, the case of  $\lambda = \mu = 0$  corresponds to "strict" inflation targeting, while "flexible" inflation targeting is characterised by  $\lambda, \mu > 0$ .

<sup>&</sup>lt;sup>17</sup>It is recognised that it would be beneficial to use a welfare criterion derived as an approximation of the representative agent's utility function (see for example Rotemberg and Woodford (1997)). The weights in this approximate welfare criterion would be functions of the parameters of the structural model itself. However, to the extent that the model used in this paper is lacking full micro-foundations, a well-defined welfare criterion does not exist.

<sup>&</sup>lt;sup>18</sup>The historical structural shocks differ from the single-equation estimation residuals, because expectations of future variables are computed to be consistent with the complete model, including the empirical benchmark for monetary policy discussed in Section 2.2. The relevant sample period is 1979:Q2 to 1998:Q4, given the estimation period for the benchmark rule.

deal with the case of interest rate rules which do not allow for a direct response to the output gap. Such rules have been used as a proxy for actual policy in inflation-targeting countries that are widely perceived as setting interest rates in response to deviations of short to medium-term inflation forecasts from the target rate.<sup>19</sup>

#### 3.2 Optimised Benchmark Rules

As a benchmark for evaluating the robustness of optimised interest rate rules under the two alternative staggered contracts models, **Table 3** reports the optimised response coefficients for a collection of outcome and forecast-based rules, together with an indication of the stabilisation performance of these rules. Regarding the choice of the forecast-based rules, we consider three different combinations of forecast horizons: one-quarter-ahead and fourquarter-ahead inflation forecasts combined with the current output gap, as used in many theoretical and empirical studies in the literature; and four-quarter-ahead forecasts of both inflation and the output gap. The last combination is motivated by **Figure 1** above which shows that a particular interest rate rule may lead to quite distinct profiles for the time paths of inflation and the output gap under the two different types of staggered contracts. As a result, choosing the forecast horizons for both inflation and the output gap may have important consequences for the stabilisation performance and robustness of interest rate rules.

The four columns in the middle of **Table 3** show the short-run response coefficients and a measure indicating the stabilisation performance of the optimised rules under Taylor-type contracts, while the four columns on the right show the corresponding results for Fuhrer-Moore-type contracts. Regardless of the policy-maker's preference for output stabilisation, we observe that the optimised rules are characterised by a substantial degree of interest rate smoothing under both types of staggered contracts, as indicated by the high coefficient on the lagged interest rate  $\rho$ . Interestingly, with the forecast horizons extending one year into the future, the magnitude of  $\rho$  tends to exceed unity, a feature which is known as

<sup>&</sup>lt;sup>19</sup>See Batini and Nelson (2001) for a model-based analysis of choosing the optimal forecast horizon within an inflation-targeting framework.

				Taylor contracts			Fuhrer-Moore contracts			
heta	$\kappa$	λ	ρ	$\alpha$	eta	$\%\Delta \mathcal{L}_{\mathrm{T}}$	ρ	$\alpha$	eta	$\%\Delta \mathcal{L}_{\mathrm{FM}}$
0	0	0	0.97	0.21	0.06	2.29	0.86	0.46	0.41	1.98
		1/2	0.86	0.03	0.40	3.62	0.85	0.41	0.53	1.80
		1	0.85	0.04	0.57	3.35	0.83	0.39	0.62	2.01
		2	0.85	0.06	0.78	3.42	0.82	0.36	0.78	2.56
1	0	0	0.98	0.23	0.05	2.90	0.89	0.45	0.35	2.16
		1/2	0.86	0.03	0.40	3.83	0.87	0.40	0.46	1.94
		1	0.85	0.04	0.56	3.36	0.86	0.38	0.55	2.19
		2	0.84	0.06	0.78	3.27	0.84	0.35	0.70	2.85
4	0	0	1.27	1.32	-0.09	5.47	0.99	0.77	0.20	4.96
		1/2	0.86	0.19	0.40	4.48	0.96	0.65	0.30	3.99
		1	0.84	0.11	0.57	3.38	0.93	0.58	0.39	4.02
		2	0.83	0.07	0.78	2.78	0.91	0.52	0.54	4.58
4	4	0	1.21	1.20	-0.09	5.58	1.07	1.00	0.37	4.92
		1/2	0.98	0.37	0.72	4.65	1.06	0.96	0.62	3.86
		1	1.00	0.33	1.21	3.80	1.07	0.96	0.87	3.82
		2	1.04	0.33	2.01	3.47	1.08	1.00	1.37	4.30

Table 3: The Stabilisation Performance of Optimised Interest Rate Rules

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the optimised interest rate response coefficients ( $\rho$ ,  $\alpha$  and  $\beta$ ) and the percentage point change in the policy-maker's loss function ( $(\Delta \mathcal{L}_i)$  compared with the loss under the fully optimal policy under commitment.

"super-inertia" in the interest rate. Not surprisingly, as the weight on output stabilisation  $\lambda$  increases, the coefficient on the output gap  $\beta$  rises while the short-run coefficient on the inflation gap  $\alpha$  falls, albeit less dramatically for Fuhrer-Moore-type contracts. Furthermore, the coefficients under Fuhrer-Moore-type contracts are typically a good bit larger than under Taylor-type contracts. This reflects that, under Fuhrer-Moore contracts, the inflation process is much harder to control and that, as a result, the policy-maker has to respond more aggressively to any signs of rising inflation.

The stabilisation performance of the optimised interest rate rules is measured in relative terms as the percentage point change in the policy-maker's loss function,  $\%\Delta\mathcal{L}_j$  (j = T, FM), compared with the loss under the fully optimal policy under commitment.<sup>20</sup> Overall, we observe that the stabilisation performance of monetary policy does not deteriorate much when the policy-maker follows simple interest rate rules rather than fully optimal policies under commitment. Under both Taylor and Fuhrer-Moore contracts the value of the policy-maker's loss function never rises by more than 6 percent. Interestingly, there is no stabilisation gain from following forecast-based as opposed to outcome-based rules.<sup>21</sup>

Based on these results, one might conclude that relying on optimised interest rate rules rather than optimal policies under commitment does not compromise the overall stabilisation performance of monetary policy significantly and, hence, seems innocuous. However, our analysis thus far has assumed that the policy-maker knows the true model of inflation determination when designing and implementing simple interest rate rules. As we will see below, the stabilisation performance of these rules can deteriorate dramatically, if this assumption is invalid.

### 4 The Robustness of Optimised Monetary Policy Rules

In the previous section it was assumed that the policy-maker knows the "true" model of inflation determination as represented by either Taylor or Fuhrer-Moore contracts. For each of these two contracting specifications, we designed optimal interest rate rules which performed remarkably well in stabilising inflation and the output gap for given preferences of the policy-maker. In the case that the optimised rules prescribed to set the interest rate in response to forecasts of future inflation or the output gap, these forecasts happened to be consistent with the structure of the model.

<sup>&</sup>lt;sup>20</sup>See Finan and Tetlow (1999) for details on computing the optimal policy under commitment for large rational expectations models using AIM.

<sup>&</sup>lt;sup>21</sup>In the sensitivity analysis at the end of the paper we consider forecast-based rules which do not allow for a direct response to the output gap. For such rules, it is found that extending the forecast horizon for inflation beyond one year leads to an improved stabilisation performance relative to outcome-based rules or rules that are confined to short inflation forecast horizons.

In the presence of uncertainty about the true model of inflation determination, however, two distinct sources of policy mistakes can be identified when implementing and designing interest rate rules. First, when implementing a forecast-based rule, the policy-maker may erroneously rely on forecasts which are obtained from the false model of inflation determination. And second, the policy-maker may already rely on the false model of inflation determination when designing the rule itself. We will refer to these two sources of policy mistakes as uncertainty about the forecasting model and uncertainty about the rule-generating model respectively. Of course, in the latter case the risks may even be heightened when also relying on the false forecasting model.

In the following we shall assess the robustness of optimised interest rate rules to uncertainty about the true model of inflation determination by evaluating the costs associated with these two distinct sources of policy mistakes.

#### 4.1 Uncertainty about the Forecasting Model

Table 4 characterises the stabilisation performance of the optimised forecast-based policy rules reported in Table 3 above under the assumption that the forecasts used when implementing these rules have been generated by the false model of the inflation process. Column four in Table 4 refers to the model with Taylor contracts and indicates the percentage point change in the policy-maker's loss function when interest rates are set according to a rule which has been optimised under Taylor contracts (representing the true model of the inflation process) but implemented using inconsistent forecasts obtained from the model with Fuhrer-Moore contracts. Here, the comparison is made with respect to the loss under the optimised forecast-based rule that has been implemented with consistent forecasts. Vice versa, column five refers to the model with Fuhrer-Moore contracts and reports the relative loss when the correctly designed policy rule is implemented using inconsistent forecasts based on the model with Taylor contracts. In each case, the inconsistent forecast,  $E_t^j[\cdot]$  (j = T, FM), is computed under the forecast-based rule designed for the model which is supposed to be the true representation of the inflation process, i.e. the policy-maker happens

			Inconsiste	nt forecasts	Pooled	forecasts
θ	$\kappa$	$\lambda$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$
1	0	0	0.26	0.66	0.08	0.16
		1/2	0.09	0.55	0.09	0.13
		1	0.15	0.53	0.15	0.13
		2	0.19	0.55	0.19	0.12
4	0	0	20.21	70.87	3.07	6.89
		1/2	4.56	61.94	2.31	5.85
		1	3.13	69.43	1.95	5.98
		2	2.37	95.51	1.72	6.81
4	4	0	22.03	61.55	3.86	6.24
		1/2	4.25	42.15	1.20	5.27
		1	2.98	35.97	0.74	5.13
		2	2.28	29.80	0.42	5.19

Table 4: The Robustness of Optimised Rules when the Forecasting Model Is Uncertain

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function ( $\%\Delta \mathcal{L}_j$ ) when the optimised forecast-based rule is implemented using either model-inconsistent or pooled forecasts, compared with the loss under the optimised forecast-based rule implemented with model-consistent forecasts.

to choose the correct rule-generating model, but when implementing the optimal rule the policy-maker resorts to the incorrect forecasting model.

The comparison of the outcomes reported in columns four and five of **Table 4** reveals that the deterioration of monetary policy is less severe, if the policy-maker overestimates the degree of inflation persistence when implementing forecast-based policy rules (that is, if forecasts that are based on Fuhrer-Moore contracts are incorrectly used in the model with Taylor contracts.) In this case, the use of inconsistent forecasts leads to a percentage point increase in the policy-maker's loss function of up to 22 percent, depending on the length of the forecast horizons and the policy-maker's preferences. In contrast, if the policy-maker underestimates the degree of inflation persistence (that is, if inflation forecasts based on Taylor contracts are erroneously implemented in the model with Fuhrer-Moore contracts), the deterioration in the performance of monetary policy is more dramatic. In this case, the loss increases by up to 71 percent. Overall, the deterioration is increasing in the length of the inflation-forecast horizon, and it is found to be particularly large if the monetary policy-maker puts no weight on output stabilisation. Interestingly, the deterioration in the performance of forecast-based rules with forecast horizons extending farther into the future seems to be alleviated at least somewhat if the horizon of the output gap forecast is synchronised with that of the inflation forecast.

The remaining two columns in **Table 4** provide information on the stabilisation performance of optimised forecast-based rules if the policy-maker implements "pooled" forecasts, which are defined as a (simple) weighted average of the individual forecasts obtained under the two alternative wage contracting specifications,  $\bar{E}_t[\cdot] = 1/2 (E_t^T[\cdot] + E_t^{FM}[\cdot])$ . In this case, the percentage point increase in the policy-maker's loss function is typically found to be lower when compared with the percentage point increase in the loss function resulting from the use of inconsistent forecasts. The relative improvement is found to be particularly large for the model with Fuhrer-Moore contracts. Hence, as far as the choice of the forecasting model is concerned, pooling forecasts rather than relying on any single model's forecast which may be largely misguiding can serve as a means to make forward-looking monetary policies more robust.

#### 4.2 Uncertainty about the Rule-Generating Model

**Table 5** summarises our findings regarding the robustness of optimised interest rate rules when there is uncertainty about the rule-generating model. We again consider the set of benchmark rules documented in **Table 3**. To assess the consequences of relying on the false model in the design of interest rate rules, we evaluate the stabilisation performance of interest rate rules that are optimised for either of the two models of inflation determination but then implemented in the alternative one which is supposed to represent the true model of the inflation process. The deterioration due to this policy mistake is measured by the

			Consistent forecasts		Inconsiste	Inconsistent forecasts		forecasts
θ	$\kappa$	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{\mathrm{FM}}$
0	0	0	20.96	224.30				
		1/2	8.04	70.01				
		1	7.83	83.02				
		2	7.40	86.10				
1	0	0	18.92	154.97	23.14	229.83	20.87	186.35
		1/2	6.95	82.63	10.22	ME	8.48	ME
		1	7.05	99.67	10.06	ME	8.47	ME
		2	6.86	103.44	9.53	ME	8.12	ME
4	0	0	8.93	58.94	33.23	558.68	16.44	88.10
		1/2	1.99	27.45	18.82	ME	7.44	ME
		1	3.09	47.06	18.10	ME	8.11	ME
		2	3.45	63.86	16.77	ME	8.04	ME
4	4	0	9.70	44.28	25.29	615.45	13.88	79.17
		1/2	1.49	20.41	18.92	ME	7.49	59.60
		1	2.22	30.83	17.74	ME	7.76	93.18
		2	2.62	34.95	15.93	ME	7.50	115.56

Table 5: The Robustness of Optimised Rules when the Rule-Generating Model Is Uncertain

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function ( $\%\Delta\mathcal{L}_j$ ) under the rule optimised for the "false" model, compared with the loss under the rule optimised for the "true" model. Each of the forecast-based rules optimised for the false model is implemented using consistent, inconsistent and pooled forecasts respectively. The notation "ME" indicates that the implemented rule yields multiple equilibria.

relative increase in the policy-maker's loss function compared with the loss that would occur under the correctly designed rule.

Starting with the optimised outcome-based rule in columns four and five of **Table 5**  $(\theta = \kappa = 0)$ , we observe that the stabilisation performance deteriorates significantly under both types of wages contracts, with the performance being distorted most severely if the policy-maker puts zero weight on output stabilisation. However, as shown in column four,

if outcome-based rules that are optimised under the assumption that the degree of inflation persistence is high (i.e. under Fuhrer-Moore contracts) are used in the low-persistence model (i.e. under Taylor contracts), the value of the loss function never rises by more than 21 percent. In contrast, as shown in column five, using outcome-based rules optimised under Taylor contracts within a model incorporating Fuhrer-Moore contracts results in distortions that are dramatically larger. In the extreme case, the value of the loss function is found to increase by about 225 percent.

We next turn to the forecast-based interest rate rules ( $\theta > 0, \kappa \ge 0$ ). Here, we consider three alternative assumptions regarding the formation of forecasts, as in Section 4.1 above. First, we assume that the policy-maker uses forecasts obtained from the correct model of the inflation process. In this case, the forecasts are model-consistent but the implemented rule is poorly designed. As indicated in columns four and five of **Table 5**, using forecastbased rules that are optimised under the wrong assumption regarding the degree of inflation persistence results in a significant deterioration in the stabilisation performance of monetary policy, as has already been documented for the outcome-based rules above. Interestingly, the deterioration is found to diminish with the length of the inflation-forecast horizons. For the model with Taylor contracts, for example, the increase in the value of the loss function amounts to 19 percent if one-quarter-ahead inflation forecasts are used, while the loss rises by less than 9 percent when using one-year-ahead inflation forecasts. Comparing the results for the two distinct contracting specifications, we observe that the relative increase in the value of the loss function for the model with Fuhrer-Moore contracts is larger by an order of magnitude.

Second, we assume that the policy-maker not only follows a poorly designed rule but also relies on model-inconsistent forecasts when implementing this rule. By comparing the results in columns four and six of **Table 5**, we observe that the deterioration in the stabilisation performance of monetary policy continues to be relatively benign if forecastbased rules optimised under Fuhrer-Moore contracts are implemented in the model with Taylor contracts using forecasts that are based on Fuhrer-Moore contracts by mistake. In contrast, as shown in column six, for the model incorporating Fuhrer-Moore contracts, implementing poorly designed rules together with inconsistent forecasts results in a dramatic deterioration of the stabilisation performance of policy. The value of the loss function increases dramatically. Even worse, in most cases the solution of the model turns out to be indeterminate, with non-fundamental shocks eventually contributing to the variance of economic fluctuations themselves.

Finally, the results reported in the last two columns of **Table 5** confirm that forecast pooling can in principle help to alleviate the distortions in the stabilisation performance of forecast-based rules, at least to the extent that those distortions arise from the use of inconsistent forecasts. The improvement however seems limited, in particular for the model with Fuhrer-Moore contracts, that is, when the true degree of inflation persistence is underestimated.

Overall, these results indicate that a cautious policy-maker who tries to avoid very poor outcomes is well-advised to design and implement monetary policies under the assumption that the degree of inflation persistence is substantial, as long as there is uncertainty regarding the true characteristics of the inflation process.

### 5 Designing Robust Monetary Policy Rules

Having documented the potential lack of robustness of optimised interest rate rules which rely on the assumption that either Taylor or Fuhrer-Moore-type staggered contracts correctly represent the inflation process, we finally proceed to identify the operating characteristics of interest rate rules which perform reasonably well under both types of staggered contracts.

In search of such robust policy rules we follow LWW (1999, 2003) and optimise the response coefficients of the parametric family of interest rate rules defined in Section 3.1 across the two contracting models simultaneously by minimising the (simple) weighted

θ	$\kappa$	λ	ρ	α	$\beta$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$
0	0	0	0.87	0.38	0.34	18.17	3.07
		1/2	0.86	0.33	0.48	9.06	2.44
		1	0.85	0.31	0.59	8.82	2.55
		2	0.84	0.30	0.76	8.62	3.03
1	0	0	0.90	0.39	0.29	17.59	3.11
		1/2	0.88	0.34	0.43	8.86	2.43
		1	0.86	0.32	0.54	8.61	2.63
		2	0.85	0.31	0.71	8.31	3.25
4	0	0	0.99	0.78	0.13	12.09	5.52
		1/2	0.94	0.63	0.31	6.03	4.11
		1	0.92	0.58	0.43	5.50	4.27
		2	0.89	0.53	0.61	4.77	4.93
4	4	0	1.03	0.91	0.21	12.56	5.59
		1/2	1.05	0.92	0.61	5.93	3.93
		1	1.06	0.95	0.94	5.60	3.94
		2	1.08	1.04	1.57	5.35	4.49

Table 6: The Stabilisation Performance of Robustly Optimised Interest Rate Rules

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ) and for each preference parameter ( $\lambda$ ), this table indicates the jointly optimised interest rate response coefficients ( $\rho$ ,  $\alpha$  and  $\beta$ ) and the percentage point change in the contributions of the alternative contracting models to the policy-maker's overall loss function ( $(\Delta \mathcal{L}_j)$ ) compared with the losses under the fully optimal policy under commitment for those models.

average of the associated loss functions,

$$ar{\mathcal{L}} ~=~ rac{1}{2} \left( \, \mathcal{L}_{\mathrm{T}} + \mathcal{L}_{\mathrm{FM}} \, 
ight).$$

Implicitly, the average loss function corresponds to the policy-maker's expected loss function when he has flat prior beliefs regarding which of these two models is the correct representation of the inflation process.

 Table 6 reports the response coefficients of the jointly, i.e. robustly optimised interest

 rate rules and indicates the stabilisation performance of these rules in terms of the percent 

age point change in the loss functions that are associated with the two distinct staggered contracts models. Again, the comparison is made with the losses under the fully optimal policies under commitment. As shown in columns four to six of **Table 6**, the size of the response coefficients of the robustly optimised rules is quite similar to the size of the optimised response coefficients that were obtained for the model with Fuhrer-Moore contracts separately (see **Table 3** above). In particular, the degree of interest rate inertia is relatively high, and the short-run response coefficients, notably those on inflation, turn out to be relatively large.<sup>22</sup> As indicated in the final two columns of **Table 6**, the loss under Fuhrer-Moore contracts increases only slightly when implementing the robustly optimised rules, while the rise in the loss under Taylor contracts is somewhat higher, in particular, if the weight on output stabilisation is zero, although still benign. Obviously, this contrasts favourably with the lack of robustness of rules that have been optimised for a particular model of the inflation process (see **Table 5** above). Importantly, none of the robustly optimised rules yields indeterminate equilibria.

Table 7 indicates the stabilisation performance of the forecast-based versions of the robustly optimised rules reported in Table 6 when these rules are implemented using either inconsistent or pooled forecasts. Here, the value of the policy-maker's loss function under the robustly optimised rule implemented with consistent forecasts is used as the benchmark for comparison again. Overall, the distortions due to the use of inconsistent forecasts are noticeable, in particular if the forecast horizon extends one year into the future. The results, however, still compare favourably with those reported in Table 4, that is when relying on rules that are optimised for the models separately. Notice, though, that using forecasts based on Taylor contracts yields indeterminate equilibria in the model with Fuhrer-Moore contracts if the policy-maker places either a very high weight on output stabilisation or none at all.

 $<sup>^{22}</sup>$ This is consistent with the findings in a study by Söderström (2002) who shows, by applying formal Bayesian analysis to a simplified version of the Rudebusch-Svensson (1999) model, that heightened responsiveness is optimal when there is uncertainty about the parameters characterising the degree of inflation persistence. The intuition is that a more aggressive response reduces uncertainty regarding the future path of inflation.

			Inconsiste	nt forecasts	Pooled forecasts		
θ	$\kappa$	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{\mathrm{FM}}$	
1	0	0	2.83	1.76	1.31	0.69	
		1/2	2.37	1.54	1.11	0.60	
		1	2.26	1.54	1.07	0.60	
		2	2.10	1.55	1.00	0.58	
4	0	0	10.84	ME	7.42	12.62	
		1/2	16.93	77.46	5.55	6.37	
		1	15.64	100.55	5.34	6.98	
		2	14.37	ME	5.02	8.17	
4	4	0	22.89	116.66	7.30	10.86	
		1/2	16.69	54.04	5.72	6.02	
		1	15.09	55.66	5.39	6.10	
		2	13.18	56.76	4.84	6.25	

Table 7: The Stabilisation Performance of Robustly Optimised Rules when the ForecastingModel Is Uncertain

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function ( $\%\Delta \mathcal{L}_j$ ) when the robustly optimised forecast-based rule is implemented using either model-inconsistent or pooled forecasts, compared with the loss under the robustly optimised rule implemented with model-consistent forecasts. The notation "ME" indicates that the implemented rule yields multiple equilibria.

As is evident from **Table 6**, one notable feature of the robustly optimised rules is the rather high degree of interest rate inertia, in particular if the interest rate is set in response to one-year-ahead inflation and/or output gap forecasts. This raises the possibility that first-difference rules that relate changes in the interest rate to inflation and output gap forecasts which do not extend too far into the future may already go a long way towards making policy rules robust to different degrees of inflation persistence.

 Table 8 summarises the stabilisation performance of a first-difference rule which relates

 the change in the short-term nominal interest rate to the one-year-ahead forecast of annual

	Consistent forecasts		Inconsiste	nt forecasts	Pooled forecasts	
$\lambda$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(1)}$	$\%\Delta {\cal L}_{ m FM}^{(1)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$
0	17.11	5.20	20.63	69.34	6.20	6.98
1/2	9.00	5.60	17.16	51.73	5.44	3.95
1	14.58	9.23	15.23	40.68	5.02	2.04
2	26.33	18.40	13.15	27.58	4.56	-0.21

 Table 8: The Stabilisation Performance of the Calibrated Forecast-Based First-Difference

 Rule

Notes: For each preference parameter  $(\lambda)$  and each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function  $(\%\Delta \mathcal{L}_j)$  under the calibrated forecast-based first-difference rule. The first-difference rule is implemented using consistent, inconsistent and pooled forecasts respectively. The superscript "(1)" indicates the comparison with the loss under the fully optimal policy under commitment, whereas the superscript "(2)" indicates the comparison with the loss under the first-difference rule implemented with consistent forecasts.

average inflation and the current output gap,

$$\Delta i_t^s = 0.75 \,\mathrm{E}_t [\,\tilde{\pi}_{t+4} - \pi^*\,] + 0.25 \,y_t,$$

with the response to the expected inflation gap calibrated to be somewhat stronger than to the current output gap.

Evidently, the calibrated first-difference rule performs remarkably well across the two models of inflation determination, in particular when the policy-maker puts a modest weight on output stabilisation. Also, the implementation of inconsistent forecasts always yields determinate solutions, with pooled forecasts providing insurance against the adverse consequences of choosing the incorrect forecasting model.

### 6 Sensitivity Analysis

Now we briefly summarise some additional sensitivity analysis regarding the results presented above. First, we consider the implications of changing the weight  $\mu$  on the variability of interest rate changes in the policy-maker's loss function. For the preceding analysis we
have chosen a weight of  $\mu = 1.0$  which, as noted in Section 3.1, implies that the variability of interest rate changes under the optimised simple rules is of the same order of magnitude as the variability implied by the estimated benchmark rule. As shown in **Table A-1** in the appendix, with a weight of  $\mu = 0.1$  on interest rate variability, the relative stabilisation performance of optimised rules is not significantly affected when compared to the baseline results reported in **Table 3** above. Similarly, we observe by comparing **Table A-2** in the appendix with **Table 4** that the results regarding the robustness of optimised rules change very little if the weight on interest rate variability is lowered to  $\mu = 0.1$ . The costs of making a policy mistake when the inflation process is in reality less persistent are not as high as the costs when the inflation process turns out to be more persistent. Finally, as documented in **Table A-3**, robustly optimised interest rate rules are yet again found to perform amazingly well across the two alternative models of inflation determination.

Second, it is worthwhile to consider the stabilisation performance and robustness of interest rate rules that respond to forecasts of inflation but do not allow for an explicit response to the output gap. Such inflation-forecast-based rules have been widely used to proxy the decision-making frameworks of central banks in inflation-targeting countries that are widely perceived as setting interest rates in response to deviations of short to mediumterm inflation forecasts from the targeted rate. In these frameworks, information on the output gap is only used to the extent that it helps to form forecasts of future inflation. Table B-1 in the appendix indicates that the performance of such rules tends to improve with increasing inflation-forecast horizon. In fact, extending the length of the inflation forecast horizon to roughly two years leads to rules that perform best within the restricted class of inflation-forecast-based rules under investigation. However, the costs of relying on inflation forecasts based on the incorrect model of the inflation process is increasing with the length of the forecast horizon as well. Most importantly, for the Fuhrer-Moore model the use of inconsistent one-year-ahead and two-year-ahead forecasts yields either indeterminate equilibria or explosive solution paths, regardless of the weight on output stabilisation. As documented in **Table B-2**, optimised rules which only respond to inflation forecasts

also lack robustness in the presence of uncertainty regarding the correct model of the inflation process. Surprisingly, though, they tend to be less prone to yielding indeterminate equilibria if the degree of inflation persistence is underestimated. Finally, as shown in **Table B-3**, optimising inflation-forecast-based rules jointly across the two models of inflation determination again helps to make them more robust, as long as the forecast horizon does not extend too far into the future and the weight on output stabilisation is relatively low.

## 7 Conclusion

This paper examined the robustness of simple monetary policy rules to the different degrees of inflation persistence generated by two distinct staggered contracts specifications within an estimated small-scale macroeconomic model of the euro area. Our central conclusion is that rules assuming a high degree of inflation persistence are more robust. More specifically, we find that it may be dangerous to rely too heavily on rules that are designed and/or implemented under the assumption that inflation persistence is low. These rules may result in disastrous stabilisation outcomes if the inflation process turns out to be considerably more persistent. In contrast, rules designed and/or implemented under the assumption that the degree of inflation persistence is relatively high also perform reasonably well if inflation persistence turns out to be low. Hence, a cautious monetary policy-maker is well-advised to take monetary policy decision under the assumption that the economy is characterised by a substantial degree of inflation persistence until strong evidence in favour of a low-inflation regime has emerged. In this context, we also find that using pooled forecasts rather than relying on a single model's forecast can serve as a means to insure the policy-maker against risks arising from the use of the wrong forecasting model when there is uncertainty on how to model the inflation process.

Regarding the key characteristics of simple monetary policy rules that are designed to be robust to different degrees of inflation persistence, we confirm earlier findings by LWW: robust rules respond to inflation and output gap forecasts with a horizon that does not extend too far in the future. Furthermore, such rules also incorporate a substantial degree of interest rate inertia. Thus, first-difference rules which relate changes in the short-term nominal interest rate to medium-term forecasts of inflation and the current output gap may already go a long way towards making monetary policy robust to uncertainty regarding the degree of inflation persistence. To the extent that such rules are found to perform remarkably well under both types of staggered contracts specifications and given that the uncertainty about inflation determination in the euro area looms large, we tentatively conclude that they may serve as a useful benchmark for model-based evaluations of monetary policy in the euro area.

There are several directions in which the analysis presented in this paper could be extended. First, while this paper has focused on the robustness of optimised interest rate rules, it would be interesting to also investigate the robustness of the fully optimal policies under commitment which were solely used as a benchmark for evaluating the stabilisation performance of optimised interest rate rules. A first attempt in this direction has been undertaken in Angeloni, Coenen and Smets (2003), though a more rigorous analysis using techniques developed in Giannoni and Woodford (2002) is left for future research. Second, in the light of the good performance of first-difference rules under both types of staggered contracts, it would be interesting to compare the robustness of rules which target the price level instead of the inflation rate, or a combination of both. Using simpler theoretical frameworks, initial studies relevant to this have been provided by Jääskelä (2002) and Batini and Yates (2002). Third, one could approach the robustness analysis using different methodologies. For example, one could design interest rate rules using an explicit minimax criterion to avoid the worst possible outcome under either of the two staggered contracts specifications. At least implicitly, part of the analysis in this paper has been following this idea, and one may form a conjecture that a more formal analysis along this line would also lead to favouring Fuhrer-Moore contracts over Taylor contracts. An example of a formal analysis using the minimax approach is Giannoni (2002) who uses a stylised New-Keynesian model though. Last, but not least, it would be important to check if the central conclusion from the paper is sensitive to alternative models of inflation determination. This could essentially be achieved by replicating the same kind of analysis using a larger set of empirical models of the inflation process. The analysis presented in Angeloni, Coenen and Smets (2003) is encouraging in this respect.

## References

- Amato, J.D. and T. Laubach, 2000, "Monetary Policy in an Estimated Optimization-Based Model with Sticky Prices and Wages", BIS Working Paper No. 87, Bank for International Settlements, May.
- Angeloni, I., G. Coenen and F. Smets, 2003, "Persistence, the Transmission Mechanism and Robust Monetary Policy", mimeo, European Central Bank, July.
- Anderson, G.S. and G.R. Moore, 1985, "A Linear Algebraic Procedure for Solving Linear Perfect Foresight models", *Economics Letters*, 17, 247-252.
- Batini, N. and E. Nelson, 2001, "Optimal Horizons for Inflation Targeting", Journal of Economic Dynamics and Control, 25 891-910.
- Batini, N. and T. Yates, 2002, "Hybrid Price Level and Inflation Targeting", forthcoming in the *Journal of Money, Credit and Banking*.
- Blanchard, O.J. and C. Kahn, 1980, "The Solution of Linear Difference Models Under Rational Expectations", *Econometrica*, 48, 1305-1311.
- Brayton, F., J.M. Roberts and J.C. Williams, 1999, "What's Happened to the Phillips Curve?", Finance and Economics Discussion Series, 99-49, Board of Governors of the Federal Reserve System, September.
- Brayton, F. and P.A. Tinsley, 1996, "A Guide to FRB/US: A Macroeconomic Model of the United States", Finance and Economics Discussion Series, 96-42, Board of Governors of the Federal Reserve System, October.
- Bryant, R.C., P. Hooper and C. Mann (eds), 1993, Evaluating Policy Regimes: New Research in Empirical Macroeconomics, Brookings Institution, Washington, DC.
- Buiter, W.H. and I. Jewitt, 1981, "Staggered Wage Setting with Real Wage Relativities: Variations on a Theme of Taylor", *The Manchester School*, 49, 211-228, reprinted in W.H. Buiter (ed.), 1989, *Macroeconomic Theory and Stabilization Policy*, Manchester University Press, Manchester
- Calvo, G.A., 1983, "Staggered Prices in a Utility-Maximizing Framework", Journal of Monetary Economics, 12, 383-398.
- Chari, V.V., P. Kehoe and E. McGrattan, 2000, "Sticky Price Models of the Business Cycle: Can the Contract Multiplier Solve the Persistence Problem?", *Econometrica*, 68, 1151-1180.
- Christiano, L.J., M. Eichenbaum and C. Evans, 2001, "Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy", NBER Working Paper No. W8403, National Bureau of Economic Research, July.
- Clarida, R., and M. Gertler, 1997, "How the Bundesbank Conducts Monetary Policy", inC. Romer and D. Romer (eds), *Reducing inflation*, NBER and University of Chicago

Press, Chicago.

- Clarida, R., J. Galí and M. Gertler, 1998, "Monetary Policy Rules in Practice: Some International Evidence", *European Economic Review*, 42, 1033-1067.
- Coenen, G. and V. Wieland, 2003, "A Small Estimated Euro Area Model with Rational Expectations and Nominal Rigidities", *European Economic Review*, forthcoming.
- Cogley, T. and T. Sargent, 2001, "Evolving Post-World War II U.S. Inflation Dynamics", in B.S. Bernanke, and J.J. Rotemberg (eds), NBER Macroeconomics Annual 2001, Vol. 16, MIT Press, Cambridge, MA.
- Erceg, C., D. Henderson, and A. Levin, 2000, "Optimal Monetary Policy with Staggered Wage and Price Contracts", *Journal of Monetary Economics*, 46, 281-313.
- Fagan, G., J. Henry and R. Mestre, 2001, "An Area-Wide Model (AWM) for the Euro Area", ECB Working Paper No. 42, European Central Bank, January.
- Faust, S., J.H. Rogers and J.H. Wright, 2001, "An Empirical Comparison of Bundesbank and ECB Monetary Policy Rules", International Finance Discussion Papers, 01-705, Board of Governors of the Federal Reserve System, August.
- Finan, F.S. and R.J. Tetlow, 1999, "Optimal Control of Large, Forward-Looking Models: Efficient Solutions and Two Examples", Finance and Economics Discussion Series, 99-51, Board of Governors of the Federal Reserve System, October.
- Fuhrer, J.C., 1997, "Inflation/Output Variance Trade-Offs and Optimal Monetary Policy", Journal of Money, Credit and Banking, 29, 214-234.
- Fuhrer, J.C. and G.R. Moore, 1995, "Inflation Persistence", Quarterly Journal of Economics, 110, 127-160.
- Galí, J. and M. Gertler, 1999, "Inflation Dynamics: A Structural Econometric Analysis", Journal of Monetary Economics, 44, 195-222.
- Galí, J., M. Gertler and J.D. López-Salido, 2001, "European Inflation Dynamics", European Economic Review, 45, 1237-1270.
- Giannoni, M.P., 2002, "Does Model Uncertainty Justify Caution? Robust Monetary Policy in a Forward-Looking Model", *Macroeconomics Dynamics*, 6, 111-144.
- Giannoni, M.P. and M. Woodford, 2002, "Optimal Interest Rates: I. General Theory", NBER Working Paper No. 9419, December.
- Guerrieri, L., 2002, "The Inflation Persistence of Staggered Contracts", International Finance Discussion Papers, 02-734, Board of Governors of the Federal Reserve System, August.
- Jääskelä, J.P., 2002, "Hybrid Inflation and Price-Level Targeting under Inflation Uncertainty", mimeo, Birkbeck College, University of London, March.

- Hansen, L.P. and T.J. Sargent, 2002, *Robust Control and Model Uncertainty in Macroeconomics*", unpublished book manuscript, University of Chicago and Hoover Institution.
- King, R. and A. Wolman, 1999, "What Should the Monetary Authority Do When Prices Are Sticky", in Taylor, 1999.
- Levin, A., V. Wieland and J.C. Williams, 1999, "Robustness of Simple Policy Rules under Model Uncertainty", in Taylor, 1999.
- Levin, A., V. Wieland and J.C. Williams, 2003, "The Performance of Forecast-Based Monetary Policy Rules under Model Uncertainty", American Economic Review, 93, 622-645.
- McCallum, B.T. and E. Nelson, 1999, "Performance of Operational Policy Rules in an Estimated Semiclassical Structural Model", in Taylor, 1999.
- OECD, 2002, OECD Economic Outlook, No. 71, Organisation for Economic Co-Operation and Development, June.
- Onatski, A. and J.H. Stock, 2002, "Robust Monetary Policy under Model Uncertainty in a Small Model of the U.S. Economy", *Macroeconomic Dynamics*, 6 85-110.
- Onatski, A. and N. Williams, 2003, "Modelling Model Uncertainty", mimeo, Columbia University and Princeton University, February
- Orphanides, A. and V. Wieland, 1998, "Price Stability and Monetary Policy Effectiveness When Nominal Interest Rates Are Bounded at Zero", Finance and Economics Discussion Series, 98-35, Board of Governors of the Federal Reserve System, June.
- Rotemberg, J. and M. Woodford, 1997, "An Optimization-Based Econometric Framework for the Evaluation of Monetary Policy", in B.S. Bernanke and J.J. Rotemberg (eds), *NBER Macroeconomics Annual*, Vol. 12, MIT Press, Cambridge, MA.
- Rotemberg, J. and M. Woodford, 1999, "Interest-Rate Rules in an Estimated Sticky Price Model", in Taylor, 1999.
- Rudd, J. and K. Whelan, 2002, "Should Monetary Policy Taget Labor's Share of Income?", mimeo, Board of Governors of the Federal Reserve System, February.
- Rudebusch, G. and L.E.O. Svensson, 1999, "Policy Rules for Inflation Targeting", in Taylor, 1999.
- Sbordone, A.M., 2002, "Price and Unit Labor Costs: A New Test of Price Stickiness", Journal of Monetary Economics, 49 265-292.
- Söderström, U., 2002, "Monetary Policy with Uncertain Parameters", Scandinavian Journal of Economics, 104 125-145.
- Staiger, D., J.H. Stock and M.W. Watson, 2001, "Prices, Wages and the U.S. NAIRU in the 1990s", mimeo, Harvard University, June.

- Svensson, L.E.O., 1999, "Inflation Targeting: Some Extensions", Scandinavian Journal of Economics, 101, 337-361.
- Taylor, J.B., 1980, "Aggregate Dynamics and Staggered Contracts", Journal of Political Economy, 88, 1-24.
- Taylor, J.B., 1993a, Macroeconomic Policy in the World Economy: From Econometric Design to Practical Operation, W.W. Norton, New York.
- Taylor, J.B., 1993b, "Discretion Versus Policy Rules in Practice", Carnegie-Rochester Conference Series on Public Policy, 39, 195-214.
- Taylor, J.B. (ed.), 1999, Monetary Policy Rules, NBER and University of Chicago Press, Chicago.
- Taylor, J.B., 2000, "Low Inflation, Pass-Through, and the Pricing Power of Firms", European Economic Review, 44, 1389-1408.
- Tetlow, R.J., and P. von zur Muehlen, 2001, "Robust Monetary Policy with Misspecified Models: Does Model Uncertainty Always Call for Attenuated Policy", Journal of Economics Dynamics and Control, 25, 911-949.
- Woodford, M., 2000, "Pitfalls of Forward-Looking Monetary Policy", American Economic Review, 90, 100-104.

			Consistent forecasts		Inconsistent forecasts		Pooled forecasts	
θ	$\kappa$	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(1)}$	$\%\Delta {\cal L}_{ m FM}^{(1)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$
0	0	0	1.55	1.87				
		1/2	4.41	1.57				
		1	3.74	2.33				
		2	3.29	3.61				
1	0	0	2.14	2.10	0.23	0.48	0.07	0.10
		1/2	4.66	1.89	0.09	0.33	0.03	0.07
		1	3.84	2.92	0.04	0.33	0.01	0.06
		2	3.31	4.59	0.01	0.35	0.00	0.05
4	0	0	4.75	4.92	28.26	ME	10.37	4.31
		1/2	5.38	3.39	3.89	33.85	0.94	3.41
		1	4.14	4.51	2.79	48.18	0.68	4.18
		2	3.42	6.60	2.10	75.45	0.52	5.68
4	4	0	4.82	4.91	17.98	53.20	3.84	4.00
		1/2	5.50	3.17	2.07	26.19	0.46	3.36
		1	4.52	4.12	1.21	28.54	0.28	3.61
		2	4.05	5.98	0.79	27.93	0.18	3.77

Table A-1: The Stabilisation Performance of Interest Rate Rules that Are Optimised with a Lower Weight of  $\mu = 0.10$  on Interest Rate Variability

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function ( $\% \Delta \mathcal{L}_j$ ) under the optimised rule. Each forecast-based rule is implemented using consistent, inconsistent and pooled forecasts respectively. The superscript "(1)" indicates the comparison with the loss under the fully optimal policy under commitment, whereas the superscript "(2)" indicates the comparison with the loss under the optimised rule implemented with consistent forecasts. The notation "ME" indicates that the implemented rule yields multiple equilibria.

			Consisten	t forecasts	Inconsistent forecasts		Pooled	forecasts
θ	$\kappa$	$\lambda$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{\mathrm{FM}}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{\mathrm{FM}}$
0	0	0	14.66	67.37				
		1/2	8.11	53.19				
		1	9.41	106.72				
		2	9.43	285.81				
1	0	0	13.29	50.90	16.07	64.48	14.57	57.03
		1/2	7.38	58.12	10.14	75.21	8.70	65.83
		1	8.98	111.16	11.76	153.25	10.32	129.03
		2	9.08	257.93	11.75	ME	10.37	ME
4	0	0	8.14	36.87	22.50	37.88	11.66	11.88
		1/2	3.37	25.23	16.15	ME	7.92	61.54
		1	5.17	35.21	18.10	ME	10.00	91.22
		2	5.24	39.63	18.08	ME	10.04	115.63
4	4	0	9.27	38.24	23.37	94.86	12.94	41.42
		1/2	2.62	30.17	16.82	188.38	7.94	61.59
		1	4.44	47.79	18.43	308.81	9.91	91.65
		2	5.15	57.55	18.25	ME	10.41	109.62

Table A-2: The Robustness of Rules that Are Optimised with a Lower Weight of  $\mu = 0.10$ on Interest Rate Variability when the Rule-Generating Model Is Uncertain

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function ( $\%\Delta \mathcal{L}_j$ ) under the rule optimised for the "false" model, compared with the loss under the rule optimised for the "true" model. Each of the forecast-based rules optimised for the false model is implemented using consistent, inconsistent and pooled forecasts respectively. The notation "ME" indicates that the implemented rule yields multiple equilibria.

			Consisten	t forecasts	Inconsistent forecasts		Pooled fo	precasts
θ	$\kappa$	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(1)}$	$\%\Delta {\cal L}_{ m FM}^{(1)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$
0	0	0	11.48	3.24				
		1/2	9.75	2.35				
		1	10.48	3.07				
		2	10.27	4.27				
1	0	0	11.14	3.40	1.84	1.35	0.84	0.56
		1/2	9.83	2.54	1.95	1.28	0.93	0.52
		1	10.42	3.60	2.12	1.33	1.01	0.53
		2	9.94	5.23	2.15	1.33	1.03	0.53
4	0	0	10.64	5.65	13.98	33.31	1.54	5.03
		1/2	7.80	3.69	12.97	40.76	4.58	4.34
		1	7.09	5.07	13.96	58.45	5.13	5.70
		2	5.77	7.22	14.07	82.13	5.14	7.05
4	4	0	10.42	6.05	14.20	33.74	3.75	5.17
		1/2	7.72	3.32	13.07	30.05	4.85	4.45
		1	7.95	4.42	13.17	31.73	5.13	4.92
		2	7.88	6.31	12.23	29.18	4.92	4.95

Table A-3: The Stabilisation Performance of Rules that Are Robustly Optimised with a Lower Weight of  $\mu = 0.10$  on Interest Rate Variability

Notes: For each choice of the inflation and output gap forecast horizons ( $\theta$  and  $\kappa$ ), for each preference parameter ( $\lambda$ ) and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function ( $\%\Delta \mathcal{L}_j$ ) under the robustly optimised rule. Each forecast-based rule is implemented using consistent, inconsistent and pooled forecasts respectively. The superscript "(1)" indicates the comparison with the loss under the fully optimal policy under commitment, whereas the superscript "(2)" indicates the comparison with the loss under the robustly optimised rule implemented with consistent forecasts.

		Consistent forecasts		Inconsiste	Inconsistent forecasts		forecasts
θ	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(1)}$	$\%\Delta {\cal L}_{ m FM}^{(1)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$
0	0	8.66	35.50				
	1/2	56.73	43.82				
	1	85.80	50.40				
	2	122.14	61.37				
1	0	6.84	27.73	-0.37	3.15	-0.33	1.04
	1/2	47.96	35.39	-0.68	2.80	-0.58	0.92
	1	74.20	41.84	-0.51	2.75	-0.54	0.89
	2	107.59	52.13	-0.34	2.69	-0.47	0.84
4	0	6.52	8.88	17.60	$\infty$	4.67	10.39
	1/2	16.77	11.81	34.33	ME	10.80	9.56
	1	27.79	15.53	44.31	ME	14.55	9.58
	2	43.41	22.66	53.20	ME	18.03	10.43
8	0	15.44	5.68	40.95	$\infty$	-0.49	16.78
	1/2	6.10	4.48	230.02	$\infty$	63.89	14.22
	1	7.19	5.27	338.89	$\infty$	97.40	12.86
	2	10.32	8.37	459.93	$\infty$	135.77	11.90

Table B-1: The Stabilisation Performance of Interest Rate Rules that Are Optimised without Allowing for a Response to the Output Gap

Notes: For each choice of the inflation forecast horizon  $(\theta)$ , for each preference parameter  $(\lambda)$  and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function  $(\% \Delta \mathcal{L}_j)$  under the optimised rule. Each forecast-based rule is implemented using consistent, inconsistent and pooled forecasts respectively. The superscript "(1)" indicates the comparison with the loss under the fully optimal policy under commitment, whereas the superscript "(2)" indicates the comparison with the loss under the optimised rule implemented with consistent forecasts. The notation "ME" indicates that the implemented rule yields multiple equilibria; the notation " $\infty$ " indicates that the implemented rule results in instability.

		Consistent forecasts		Inconsiste	Inconsistent forecasts		forecasts
θ	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$	$\%\Delta \mathcal{L}_{\mathrm{T}}$	$\%\Delta \mathcal{L}_{ m FM}$
0	0	32.08	180.81				
	1/2	7.82	14.66				
	1	4.43	5.53				
	2	2.68	2.71				
1	0	24.88	126.26	33.46	193.07	28.76	154.11
	1/2	4.98	10.60	8.60	20.82	6.44	14.92
	1	2.23	3.25	4.30	8.98	2.90	5.56
	2	1.73	1.85	2.42	4.50	1.67	2.76
4	0	4.01	25.88	16.42	416.35	8.43	52.76
	1/2	1.47	3.56	18.96	58.45	4.82	1.79
	1	4.14	8.86	14.42	31.66	3.57	-0.38
	2	9.72	17.28	11.32	13.40	3.96	0.98
8	0	-0.80	177.03	21.65	$\infty$	1.17	9.43
	1/2	9.57	66.46	17.33	ME	3.83	12.59
	1	17.79	104.71	16.85	ME	7.08	24.40
	2	29.14	148.59	17.84	ME	12.02	39.50

Table B-2: The Robustness of Rules that Are Optimised without Allowing for a Response to the Output Gap when the Rule-Generating Model Is Uncertain

Notes: For each choice of the inflation forecast horizon  $(\theta)$ , for each preference parameter  $(\lambda)$  and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function  $(\% \Delta \mathcal{L}_j)$  under the rule optimised for the "false" model, compared with the loss under the rule optimised for the "true" model. Each of the forecast-based rules optimised for the false model is implemented using consistent, inconsistent and pooled forecasts respectively. The notation "ME" indicates that the implemented rule yields multiple equilibria; the notation " $\infty$ " indicates that the implemented rule results in instability.

		Consistent forecasts		Inconsiste	Inconsistent forecasts		forecasts
θ	λ	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(1)}$	$\%\Delta {\cal L}_{ m FM}^{(1)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$	$\%\Delta \mathcal{L}_{\mathrm{T}}^{(2)}$	$\%\Delta {\cal L}_{ m FM}^{(2)}$
0	0	31.80	37.70				
	1/2	62.63	45.11				
	1	88.85	51.41				
	2	123.96	62.19				
1	0	25.21	29.33	5.15	5.00	2.30	2.13
	1/2	51.93	36.12	2.18	4.17	0.79	1.57
	1	75.88	42.30	1.08	3.56	0.22	1.29
	2	108.77	52.63	0.09	2.70	-0.31	0.89
4	0	10.07	9.03	28.70	154.65	5.99	20.11
	1/2	17.90	11.95	19.70	150.19	4.45	7.26
	1	31.02	16.03	15.53	ME	2.11	4.70
	2	50.84	24.24	11.29	ME	-0.39	1.39
8	0	14.49	5.69	23.31	$\infty$	-5.63	16.16
	1/2	14.31	4.97	13.66	$\infty$	-2.87	9.44
	1	21.47	6.47	9.99	$\infty$	-5.05	5.83
	2	32.80	10.76	6.73	$\infty$	-7.22	1.74

Table B-3: The Stabilisation Performance of Rules that Are Robustly Optimised without Allowing for a Response to the Output Gap

Notes: For each choice of the inflation forecast horizon  $(\theta)$ , for each preference parameter  $(\lambda)$  and for each contracting specification (j), this table indicates the percentage point change in the policy-maker's loss function  $(\%\Delta \mathcal{L}_j)$  under the robustly optimised rule. Each forecast-based rule is implemented using consistent, inconsistent and pooled forecasts respectively. The superscript "(1)" indicates the comparison with the loss under the fully optimal policy under commitment, whereas the superscript "(2)" indicates the comparison with the loss under the robustly optimised rule implemented with consistent forecasts. The notation "ME" indicates that the implemented rule yields multiple equilibria; the notation " $\infty$ " indicates that the implemented rule results in instability.

## European Central Bank working paper series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.int).

- 202 "Aggregate loans to the euro area private sector" by A. Calza, M. Manrique and J. Sousa, January 2003.
- 203 "Myopic loss aversion, disappointment aversion and the equity premium puzzle" by D. Fielding and L. Stracca, January 2003.
- 204 "Asymmetric dynamics in the correlations of global equity and bond returns" by L. Cappiello, R.F. Engle and K. Sheppard, January 2003.
- 205 "Real exchange rate in an inter-temporal n-country-model with incomplete markets" by B. Mercereau, January 2003.
- 206 "Empirical estimates of reaction functions for the euro area" by D. Gerdesmeier and B. Roffia, January 2003.
- 207 "A comprehensive model on the euro overnight rate" by F. R. Würtz, January 2003.
- 208 "Do demographic changes affect risk premiums? Evidence from international data" by A. Ang and A. Maddaloni, January 2003.
- 209 "A framework for collateral risk control determination" by D. Cossin, Z. Huang, D. Aunon-Nerin and F. González, January 2003.
- 210 "Anticipated Ramsey reforms and the uniform taxation principle: the role of international financial markets" by S. Schmitt-Grohé and M. Uribe, January 2003.
- 211 "Self-control and savings" by P. Michel and J.P. Vidal, January 2003.
- 212 "Modelling the implied probability of stock market movements" by E. Glatzer and M. Scheicher, January 2003.
- 213 "Aggregation and euro area Phillips curves" by S. Fabiani and J. Morgan, February 2003.
- 214 "On the selection of forecasting models" by A. Inoue and L. Kilian, February 2003.
- 215 "Budget institutions and fiscal performance in Central and Eastern European countries" by H. Gleich, February 2003.
- 216 "The admission of accession countries to an enlarged monetary union: a tentative assessment" by M. Ca'Zorzi and R. A. De Santis, February 2003.
- 217 "The role of product market regulations in the process of structural change" by J. Messina, March 2003.

- 218 "The zero-interest-rate bound and the role of the exchange rate for monetary policy in Japan" by G. Coenen and V. Wieland, March 2003.
- 219 "Extra-euro area manufacturing import prices and exchange rate pass-through" by B. Anderton, March 2003.
- 220 "The allocation of competencies in an international union: a positive analysis" by M. Ruta, April 2003.
- 221 "Estimating risk premia in money market rates" by A. Durré, S. Evjen and R. Pilegaard, April 2003.
- 222 "Inflation dynamics and subjective expectations in the United States" by K. Adam and M. Padula, April 2003.
- 223 "Optimal monetary policy with imperfect common knowledge" by K. Adam, April 2003.
- 224 "The rise of the yen vis-à-vis the ("synthetic") euro: is it supported by economic fundamentals?" by C. Osbat, R. Rüffer and B. Schnatz, April 2003.
- 225 "Productivity and the ("synthetic") euro-dollar exchange rate" by C. Osbat, F. Vijselaar and B. Schnatz, April 2003.
- 226 "The central banker as a risk manager: quantifying and forecasting inflation risks" by L. Kilian and S. Manganelli, April 2003.
- 227 "Monetary policy in a low pass-through environment" by T. Monacelli, April 2003.
- 228 "Monetary policy shocks a nonfundamental look at the data" by M. Klaeffing, May 2003.
- 229 "How does the ECB target inflation?" by P. Surico, May 2003.
- 230 "The euro area financial system: structure, integration and policy initiatives" by P. Hartmann, A. Maddaloni and S. Manganelli, May 2003.
- 231 "Price stability and monetary policy effectiveness when nominal interest rates are bounded at zero" by G. Coenen, A. Orphanides and V. Wieland, May 2003.
- 232 "Describing the Fed's conduct with Taylor rules: is interest rate smoothing important?" by E. Castelnuovo, May 2003.
- 233 "The natural real rate of interest in the euro area" by N. Giammarioli and N. Valla, May 2003.
- 234 "Unemployment, hysteresis and transition" by M. León-Ledesma and P. McAdam, May 2003.
- 235 "Volatility of interest rates in the euro area: evidence from high frequency data" by N. Cassola and C. Morana, June 2003.

- 236 "Swiss monetary targeting 1974-1996: the role of internal policy analysis" by G. Rich, June 2003.
- 237 "Growth expectations, capital flows and international risk sharing" by O. Castrén, M. Miller and R. Stiegert, June 2003.
- 238 "The impact of monetary union on trade prices" by R. Anderton, R. E. Baldwin and D. Taglioni, June 2003.
- 239 "Temporary shocks and unavoidable transitions to a high-unemployment regime" by W. J. Denhaan, June 2003.
- 240 "Monetary policy transmission in the euro area: any changes after EMU?" by I. Angeloni and M. Ehrmann, July 2003.
- 241 Maintaining price stability under free-floating: a fearless way out of the corner?" by C. Detken and V. Gaspar, July 2003.
- 242 "Public sector efficiency: an international comparison" by A. Afonso, L. Schuknecht and V. Tanzi, July 2003.
- 243 "Pass-through of external shocks to euro area inflation" by E. Hahn, July 2003.
- 244 "How does the ECB allot liquidity in its weekly main refinancing operations? A look at the empirical evidence" by S. Ejerskov, C. Martin Moss and L. Stracca, July 2003.
- 245 "Money and payments: a modern perspective" by C. Holthausen and C. Monnet, July 2003.
- 246 "Public finances and long-term growth in Europe evidence from a panel data analysis" by D. R. de Ávila Torrijos and R. Strauch, July 2003.
- 247 "Forecasting euro area inflation: does aggregating forecasts by HICP component improve forecast accuracy?" by K. Hubrich, August 2003.
- 248 "Exchange rates and fundamentals" by C. Engel and K. D. West, August 2003.
- 249 "Trade advantages and specialisation dynamics in acceding countries" by A. Zaghini, August 2003.
- 250 "Persistence, the transmission mechanism and robust monetary policy" by I. Angeloni, G. Coenen and F. Smets, August 2003.
- 251 "Consumption, habit persistence, imperfect information and the lifetime budget constraint" by A. Willman, August 2003.
- 252 ""Interpolation and backdating with a large information set" by E. Angelini, J. Henry and M. Marcellino, August 2003.
- 253 "Bond market inflation expectations and longer-term trends in broad monetary growth and inflation in industrial countries, 1880-2001" by W. G. Dewald, September 2003.

- 254 "Forecasting real GDP: what role for narrow money?" by C. Brand, H.-E. Reimers and F. Seitz, September 2003.
- 255 "Is the demand for euro area M3 stable?" by A. Bruggeman, P. Donati and A. Warne, September 2003.
- 256 "Information acquisition and decision making in committees: a survey" by K. Gerling,H. P. Grüner, A. Kiel and E. Schulte, September 2003.
- 257 "Macroeconomic modelling of monetary policy" by M. Klaeffling, September 2003.
- 258 "Interest rate reaction functions and the Taylor rule in the euro area" by P. Gerlach-Kristen, September 2003.
- 259 "Implicit tax co-ordination under repeated policy interactions" by M. Catenaro and J.-P. Vidal, September 2003.
- 260 "Aggregation-theoretic monetary aggregation over the euro area, when countries are heterogeneous" by W. A. Barnett, September 2003.
- 261 "Why has broad money demand been more stable in the euro area than in other economies? A literature review" by A. Calza and J. Sousa, September 2003.
- 262 "Indeterminacy of rational expectations equilibria in sequential financial markets" by P. Donati, September 2003.
- 263 "Measuring contagion with a Bayesian, time-varying coefficient model" by M. Ciccarelli and A. Rebucci, September 2003.
- 264 "A monthly monetary model with banking intermediation for the euro area" by A. Bruggeman and M. Donnay, September 2003.
- 265 "New Keynesian Phillips Curves: a reassessment using euro area data" by P. McAdam and A. Willman, September 2003.
- 266 "Finance and growth in the EU: new evidence from the liberalisation and harmonisation of the banking industry" by D. Romero de Ávila, September 2003.
- 267 "Comparing economic dynamics in the EU and CEE accession countries" by R. Süppel, September 2003.
- 268 "The output composition puzzle: a difference in the monetary transmission mechanism in the euro area and the US" by I. Angeloni, A. K. Kashyap, B. Mojon and D. Terlizzese, September 2003.
- 269 "Zero lower bound: is it a problem with the euro area?" by G. Coenen, September 2003.
- 270 "Downward nominal wage rigidity and the long-run Phillips curve: simulation-based evidence for the euro area" by G. Coenen, September 2003.
- 271 "Indeterminacy and search theory" by N. Giammarioli, September 2003.

- 272 ""Inflation targets and the liquidity trap" by M. Klaeffling and V. López Pérez, September 2003.
- 273 "Definition of price stability, range and point inflation targets: the anchoring of long-term inflation expectations" by E. Castelnuovo, S. Nicoletti-Altimari and D. Rodriguez-Palenzuela, September 2003.
- 274 "Interpreting implied risk neutral densities: the role of risk premia" by P. Hördahl and D. Vestin, September 2003.
- 275 "Identifying the monetary transmission mechanism using structural breaks" by A. Beyer and R. Farmer, September 2003.
- 276 "Short-term estimates of euro area real GDP by means of monthly data" by G. Rünstler, September 2003.
- 277 "On the indeterminacy of determinacy and indeterminacy" by A. Beyer and R. Farmer, September 2003.
- 278 "Relevant economic issues concerning the optimal rate of inflation" by D. R. Palenzuela, G. Camba-Méndez and J. Á. García, September 2003.
- 279 "Designing targeting rules for international monetary policy cooperation" by G. Benigno and P. Benigno, October 2003.
- 280 "Inflation, factor substitution and growth" by R. Klump, October 2003.
- 281 "Identifying fiscal shocks and policy regimes in OECD countries" by G. de Arcangelis and S. Lamartina, October 2003.
- 282 "Optimal dynamic risk sharing when enforcement is a decision variable" by T. V. Koeppl, October 2003.
- 283 "US, Japan and the euro area: comparing business-cycle features" by P. McAdam, November 2003.
- 284 "The credibility of the monetary policy 'free lunch'" by J. Yetman, November 2003.
- 285 "Government deficits, wealth effects and the price level in an optimizing model" by B. Annicchiarico, November 2003.
- 286 "Country and sector-specific spillover effects in the euro area, the United States and Japan" by B. Kaltenhaeuser, November 2003.
- 287 "Consumer inflation expectations in Poland" by T. Łyziak, November 2003.
- 288 "Implementing optimal control cointegrated I(1) structural VAR models" by F. V. Monti, November 2003.
- 289 "Monetary and fiscal interactions in open economies" by G. Lombardo and A. Sutherland, November 2003.

290 "Inflation persistence and robust monetary policy design" by G. Coenen, November 2003.